Author's response to reviews

**Title:** Trigeminal Electrophysiology: A 2x2 matrix model for differential diagnosis between Temporomandibular Disorders and Orofacial Pain

**Authors:**

- Gianni Frisardi (frisardi@tin.it)
- Giacomo Chessa (gchessa@uniss.it)
- Gianfranco Sau (gfs@tiscali.it)
- Flavio Frisardi (frisardi.flavio@tiscali.it)

**Version:** 4  **Date:** 18 May 2010

**Author's response to reviews:** see over
May 18th, 2010

Dear Editor,

Thank you very much for your email message dated April 15th, 2010, in which you expressed a great deal of appreciation for our work and gave us suggestions for improving the manuscript entitled “Trigeminal Electrophysiology in Patients with Orofacial Pain and Temporomandibular Disorders” (MS: 4517918732514052) for possible publication in *BMC Musculoskeletal Disorders*.

As per your request and the comments of the three Reviewers, we have made changes to the manuscript, by including new three clinical cases in the discussion section and by revising both text and figures.

An itemized list of all changes made is enclosed.

Indeed, my co-authors and I believe that the manuscript has been significantly improved, and we hope that you will find it suitable for publication in *BMC Musculoskeletal Disorders*.

Sincerely,

Gianni Frisardi

Orofacial Pain Center “Epochè”
Viale G. Matteotti 91, 00048 Nettuno (Rome) -Italy. Tel: 0039 06 9804953,
Fax.: 0039 06 98832103.email: frisardi@tin.it